# Kent Academic Repository Full text document (pdf)

# **Citation for published version**

Callaghan, Lisa and Brookes, Nadia and Palmer, Sinead (2017) Older people receiving family-based support in the community: A survey of quality of life amongst users of 'Shared Lives' in England. Health & Social Care in the Community, 25 (5). pp. 1655-1666. ISSN 0966-0410.

# DOI

https://doi.org/10.1111/hsc.12422

# Link to record in KAR

http://kar.kent.ac.uk/52924/

# **Document Version**

Publisher pdf

# **Copyright & reuse**

Content in the Kent Academic Repository is made available for research purposes. Unless otherwise stated all content is protected by copyright and in the absence of an open licence (eg Creative Commons), permissions for further reuse of content should be sought from the publisher, author or other copyright holder.

# Versions of research

The version in the Kent Academic Repository may differ from the final published version. Users are advised to check http://kar.kent.ac.uk for the status of the paper. Users should always cite the published version of record.

# Enquiries

For any further enquiries regarding the licence status of this document, please contact: **researchsupport@kent.ac.uk** 

If you believe this document infringes copyright then please contact the KAR admin team with the take-down information provided at http://kar.kent.ac.uk/contact.html





Health and Social Care in the Community (2017)

# Older people receiving family-based support in the community: a survey of quality of life among users of 'Shared Lives' in England

Lisa Callaghan BSc MSc, Nadia Brookes BA MSc PGCAP and Sinead Palmer BSc Personal Social Services Research Unit, University of Kent, Canterbury, UK

Accepted for publication 13 December 2016

#### Correspondence

Lisa Callaghan Personal Social Services Research Unit, University of Kent Canterbury, Kent CT2 7NF, UK E-mail: I.a.callaghan@kent.ac.uk

#### What is known about this topic

- Shared Lives support is mainly used by people with learning disabilities.
- There is some evidence of high levels of satisfaction among users and Shared Lives is rated highly by the Care Quality Commission in England.
- Shared Lives has been identified as a potential option for older people.

#### What this paper adds

- Shared Lives appears to deliver good outcomes for older people.
- In comparison to a matched group of older people using other forms of social care, Shared Lives users reported better quality of life in some areas.
- Shared Lives should be included as part of support options offered to older people and their families.

#### Abstract

Shared Lives (adult placement) is a model of community-based support where an adult who needs support and/or accommodation moves into or regularly visits the home of an approved Shared Lives carer, after they have been matched for compatibility. It is an established but small service which has been used mainly by people with learning disabilities but which has the potential to offer an alternative to traditional services for some older people. However, there is little research on the outcomes for older users of Shared Lives. This paper presents findings from a survey of 150 older people using Shared Lives support across 10 Shared Lives schemes in England, which took place between June 2013 and January 2014. The aim was to identify outcomes for older users of Shared Lives and compare these to outcomes for older users of other social care services. In the absence of an ideal study design involving randomised allocation, statistical matching was used to generate a comparison group from the Adult Social Care Survey from 2011/12, with 121 cases matched to 121 Shared Lives cases. The main outcome measures were Social Care-Related Quality of Life (measured by the ASCOT) and overall quality of life. Findings indicated that Shared Lives can deliver good outcomes for older people, particularly for overall quality of life. In comparison to the matched group of older people using other forms of support, there was some evidence that Shared Lives may deliver better outcomes in some aspects of quality of life. Limitations to the research mean, however, that more work is needed to fully understand the role Shared Lives could play in supporting older people.

**Keywords:** community-based support, older people, personalisation, quality of life, Shared Lives

#### Introduction

At a time of substantial cuts to social care spending and an ageing population, policy makers and practitioners must ascertain how best to provide for the care and support needs of older people. Central to current social care policy in the UK and Western Europe is the personalisation agenda with an emphasis on giving people active choice and control over their care and support (DH, 2010, 2012). Additionally, in recent years, councils in England have been encouraged to reduce inappropriate admissions to residential care through improving options for community-based care and support (DH, 2010) and changing their focus to prevention and early intervention (Care Act, 2014). Older people have been identified as among those hard to include in the move to personalisation (Newbronner *et al.* 2011), and concern has been expressed regarding the effectiveness of home care services for delivering personalised care and support for this group (Equality and Human Rights Commission, 2011). The challenge of combating loneliness and social isolation, to which people can be particularly vulnerable in older age (Windle *et al.* 2011) is not easily met by these services, particularly in times of increasing budget cuts. It has been noted that as people age quality of life is particularly shaped by the relationships they have and communities they live in (Blood 2013), and that attachment to place and a sense of belonging to the physical and social environment may increase (Lawton 1985, Gilleard *et al.* 2007). In recognition of these factors, practitioners, commissioners, service users and carers need evidence on alternative, community-based service options for older people.

Scoping research funded by the National Institute for Health Research (NIHR) School for Social Care Research (SSCR) highlighted Shared Lives (previously known as adult or family placement) as a personalised service mainly used by people with learning disabilities, which may have the potential to deliver good outcomes for older people (Callaghan et al. 2012). This paper draws on a follow-on study which aimed to generate evidence about the use of Shared Lives (henceforth noted as SL) for older people. It presents findings from a national survey of older people using SL, describing quality of life outcomes. Outcomes are compared with a matched sample of older people using other forms of support. Following this, implications of the findings are discussed.

### Shared lives

Adult or family placement-type services have been used for many years with different client groups and in most parts of the world, particularly Northern Europe and the USA, although terminology and definitions vary (Schofield 2009). The oldest formally constituted service is in Geel, Belgium where people with mental health conditions and learning disabilities have been supported in local families for centuries, while in the UK the Liverpool Personal Service Society has been providing adult placements since the 1970s (Fiedler 2005). There was a growth in adult placements in the UK during the 1970s following the setting up of local authority social service departments and a policy shift away from institutional care towards care in the community (Schofield 2009), and by the 1980s many schemes mainly offered long-term care and support for people with learning disabilities (Dagnan 1997).

Today, SL is the term used to describe familybased support where service users are included in the family and community life of a 'Shared Lives carer', who uses their family home as a resource (see Box 1 for a detailed definition). There are different types of SL arrangement: residential or long

#### Box 1 Detailed definition of Shared Lives

- Arrangements are part of organised Shared Lives Schemes that approve and train Shared Lives Carers, receive referrals, match the needs of service users with Shared Lives Carers, and monitor the arrangements.
- People using Shared Lives services have the opportunity to be part of the Shared Lives Carer's family and social networks. Shared Lives Carers use their family home as a resource.
- Arrangements provide committed and consistent relationships. The relationship between the Shared Lives Carer and the
- person placed with them is of mutual benefit. Shared Lives Carers can support up to three people at any one
- time (up to two people in Wales).
- Shared Lives carers do not employ staff to provide care to the people placed with them.

Source: sharedlivesplus.org.uk

term; respite or short breaks; day support; rehabilitative or intermediate support; and outreach support (which incorporates elements of the SL model, but is delivered in the service user's home). SL carers are recruited, trained and approved by local schemes, and are paid a fixed amount rather than an hourly rate. They are carefully matched to the person requiring support to ensure compatibility, a crucial element of the SL model. The key difference between SL and other community-based models, such as Supported Living, lies in the importance placed on integration with a family and mutuality of relationships.

Until recently, the extent of the use of SL support in the UK, and by whom, was unknown. Data from a survey of SL schemes in England conducted by Shared Lives Plus (the UK network for family-based and small-scale ways of supporting adults) were used to estimate that over 9660 people were supported by SL at the end of 2013, 1600 of whom were older people. This includes older people who are receiving support from SL for other reasons such as a learning disability or physical impairment. Overall, it was projected that around 7% of the total number of people receiving SL support had dementia or were frail older people (Shared Lives Plus, 2014). As such, SL represents a small proportion of adult care and support services, although the sector is growing (Shared Lives Plus, 2016).

SL has strong advocates, and has been compared favourably to other care and support options on some key indicators of personalisation, such as inclusion, flexibility, choice and control (NAAPS, 2010). It is consistently rated highly by the Care Quality Commission (the regulator for health and social care

Shared Lives is a service provided by individuals and families in local communities and is distinguished by the following features:

reived demand for SL from older people and \*The easy-read version of the questionnaire had fewer items;

roval from the Social for the purposes of this paper, only measures included in both versions are reported. tee, support from the <sup>†©</sup>1990 EuroQol Group. EQ-5D<sup>™</sup> is a trademark of the EuroQo

<sup>1©</sup>1990 EuroQol Group. EQ-5D<sup>™</sup> is a trademark of the EuroQol Group.

services in England) and in 2013/2014 was rated as achieving 100% compliance with quality standards in relation to respect and involvement (Care Quality Commission, 2014).

There has been relatively little research on SL, although there is some evidence of high levels of satisfaction among users (Fiedler 2005, NAAPS and IESE, 2009). Research studies have suggested that users value being treated as an individual, being part of a family and taking part in household tasks (Ware 1987, Dagnan & Drewett 1988, Robinson & Simons 1996). As part of the Shared Lives Plus survey of SL schemes, a survey of 80 SL carers indicated that SL arrangements had helped users (including older people) develop independent living skills and increase their social participation (Shared Lives Plus, 2014). There has been little attempt to compare outcomes of SL support to other forms of care (Dagnan 1997, Schofield 2009). Although the majority of SL support in the UK is for people with learning disabilities, it has been identified as having potential for older people, including those with dementia (Valios 2010, Fox 2011, Bell & Litherland 2013). SL has been proposed as a possible alternative to traditional respite care (McConkey et al. 2002, Valios 2010) and may also provide an alternative to moving to a care home for some. There is some indication that SL schemes and local authorities would support expansion of their services to include more older people (Brookes & Callaghan 2013). However, there is a lack of robust evidence about outcomes for older people using SL, and how these may compare to older people using alternative care and support. To begin to address this, the SSCR commissioned the Personal Social Services Research Unit at the University of Kent to examine the potential of SL for older people.

## Methods

## The study

The findings reported here draw on a survey of older people using SL support in England. The survey was part of a wider study conducted between January 2012 and April 2014, which collected information on the outcomes and experiences of older people using SL and their family carers, issues for implementation and expansion of SL schemes to support older people, perceived demand for SL from older people and costs.

The study had ethics approval from the Social Care Research Ethics Committee, support from the Association of Directors of Adult Social Services and research governance approval from participating councils. It benefited from the guidance of national and local project advisory groups involving practitioners, SL users and carers, family carers and academics.

## Data collection

The aim of the survey was to identify the outcomes of older users of SL and compare those outcomes to older users of other social care services. In the absence of an ideal study design involving randomised allocation, statistical matching (described below) was used to generate a comparison group from the Adult Social Care Survey (ASCS) from 2011/2012.

A self-completion questionnaire was developed using questions from the ASCS and other existing surveys to facilitate comparison, alongside a number of demographic questions (see Box 2). An 'easy-read' version of the questionnaire was developed based on the easy-read ASCS, and was made available to anyone who preferred this format. The main outcome measure used was the Adult Social Care Outcomes Toolkit (ASCOT) (Netten *et al.* 2012), designed to measure aspects of quality of life specifically relating to social care and applicable across different care settings, with all user groups (http://www.pssru.ac.uk/ascot).

Box 2 Items and measures included in questionnaire\*

Background information Health and dependency	Age, gender, ethnicity Type of SL support Self-perceived health (Robine <i>et al.</i> 2002), 5-point scale Ability to perform activities of daily living
	(ADLs) and instrumental activities of daily living (IADLs), eight individual items, scores summed to give dependency score from 0–16
	Two items from the EQ5D (Euro-QoL <sup>†</sup> ) which measures health-related quality of life: extent of pain/discomfort, extent of anxiety/ depression. 3-point scale
Assistance with	Level of help (single question)
Outcome measures	Overall quality of life (Bowling 1995); 7-point scale, collapsed into 5 points to be consistent with easy-read version Social care-related quality of life (measured using the ASCOT; Netten <i>et al.</i> 2012)

The eight domains of social care-related quality of life (SCRQoL) that the ASCOT measure draws on are personal cleanliness and comfort; food and drink; accommodation cleanliness and comfort; safety; control over daily life; social participation and involvement; occupation; and dignity. Items are coded to represent the 'ideal' state, having no needs, some needs or high needs in each domain (see Box 3). To create an overall SCRQoL 'score', responses are weighted to reflect the relative importance of each domain and level of need, based on previous work on the preferences of social care users and the general population, estimated using a combination of Best-Worst Scaling and Time Trade-Off approaches (Potoglou et al. 2011, Netten et al., 2012, pp. 85-87). These are then summed to give a total ranging from -0.17 to 1.00, with 0 equating to 'being dead' and 1.00 to an 'ideal' state; scores less than 0 refer to a state that is rated as being worse than death (Netten et al., 2012).

SL schemes were recruited through a scoping survey in July 2012 and a follow-up email from Shared Lives Plus. All schemes providing services to older people were asked to volunteer, and 12 agreed. Managers of the schemes were sent documentation relating to the survey: invitation letters for participants including information about the project and what participation would involve, questionnaires and envelopes for their return. As numbers of older people using SL were expected to be small, schemes were asked to send these to all eligible users of their service, and to return completed surveys to the researchers. Following advice from the advisory groups, a broad inclusion criteria were used; older people (age 65 and over), including people with learning disabilities, and using any form of SL support (including outreach delivered in the service user's own home). Only individuals with capacity to consent and to understand the questions were included, a judgement made by the SL workers distributing the questionnaire based on their knowledge of the service users in their scheme. Schemes were sent questionnaires corresponding to the number of

Box 3 Meaning of response options

Ideal	The preferred situation, in which needs are met to the desired level
No needs	Where needs are met, but not to the desired level
Low-level needs	Where there are needs, but these do not have an immediate or long-term health implication
High-level needs	Where there are needs and these have an immediate or long-term health implication

Source: Netten et al. (2012).

4

© 2017 The Authors. Health and Social Care in the Community Published by John Wiley & Sons Ltd

eligible service users they had (ranging from 3 to over 200), a total of 430. Data collection took place between June 2013 and January 2014.

## Data analysis

Descriptive statistics were used to describe the SL sample. For comparisons between participants using different types of SL, chi-squared ( $\chi^2$ ) tests were used for variables with unordered categories. Kruskal-Wallis or Mann-Whitney tests were computed for variables with small numbers of ordered categories. When the variable was continuous or had more than five meaningfully ordered categories, t-tests or ANOVA were used. Posthoc comparisons (the Bonferroni comparison) were used where appropriate to compare differences between groups. Multivariate analyses were used to explore factors associated with variations in outcome in the SL sample. Screening of the data indicated that it did not meet a number of assumptions needed to perform ordinary least squares (OLS) multiple regression, and so regression with robust standard errors (Chen et al., 2003) was performed. Selection of variables for inclusion was based on their relationship with the outcome variable and with other variables of interest such as type of SL. The small sample size, however, meant that this was limited to a relatively small number of explanatory variables (Tabachnick & Fidell 2007). Only complete cases were included in the regression models. Just 3% of the sample was missing the overall quality of life score but 15 people were missing a SCRQoL score due to missing data on one or more items. A logistic regression model was computed using the outcome variable as a dummy variable to indicate 'missing' or 'not missing' the SCRQOL score. The model was non-significant, suggesting that those missing a SCRQoL score did not differ significantly from those who did not in terms of background characteristics, and so imputation of missing data was not deemed necessary.

Tests for difference between the SL sample and the ASCS sample were computed. Given that the two samples were different, propensity score matching (PSM) (Rosenbaum & Rubin 1983, Austin 2011) was used to generate a matched sample, adjusting for baseline imbalances in the two groups and so facilitate an unbiased comparison of outcomes. PSM is a non-experimental method of sampling that produces a 'control group' whose distribution of variables is similar to that of the study group. A significance level of 0.05 was adopted for all tests. Data were analysed using Stata SE13.

#### Results

#### Participants

The sample comprised 150 older users of SL, a response rate of 29% of the total number of questionnaires requested by schemes (it was not possible to identify how many were actually sent out) and approximately 9% of the total number of older people estimated by Shared Lives Plus to be using SL at the time of the survey.

Half of the sample were female. The age of respondents ranged from 65 to 102, with an average age of 77 (SD = 8.68). The vast majority (98%) of the sample was white British, with one Asian/Asian British and two black/black British respondents.

Ouestionnaires were returned from 10 SL schemes, with a range of 1-64 per scheme. Four 'types' of SL were represented: long-term/residential, day support, respite/short breaks and outreach support (see Table 1). Long-term SL users were spread across eight schemes, while the majority (89%) of day support users came from only one scheme. Respite was used by the smallest number of people. The outreach users made up a significant proportion of the sample and came from one scheme. Although this service is not traditionally considered as SL, the matching process remains the same, meaning that it is possible to attain the same level of personalisation as in 'traditional' SL services. Forty people completed the easyread version of the questionnaire. Thirty-six of these were using long-term SL, reflecting the tendency for long-term placements to be used by people with a learning disability.

Table 2 shows the differences between individuals using different types of SL arrangement. There was a broad range of ages across all groups. Tests indicated that those using long-term SL were younger than those using day support or outreach. They were also less dependent, reported experiencing less pain or discomfort, less anxiety and depression and rated their health as better. There was no significant difference in proportions of males and females.

Table 1	Responses	bv	type	of	SL	arrangement
Tuble I	11000011000	ωy	ype	01		unungemen

Type of arrangement	No. of schemes	No. of responses (% of sample)
Long term/residential	8	52 (34.7)
Day support	2	35 (23.3)
Respite/short breaks	3	8 (5.3)
Outreach/kinship support	1	55 (36.7)
Total		150

Those using long-term SL were less likely to have had someone else complete the answers on their behalf (by proxy), while outreach users were more likely to have used a proxy and less likely to have received no assistance. These differences need to be taken into account when describing the outcomes for older people using SL; users of different types of SL should not be treated as a homogenous group.

#### Quality of life outcomes

Overall, 74% of the older people using SL rated their quality of life as 'good' or 'very good', 22% as 'alright', and 4% as 'bad' or 'very bad'. The average SCRQoL score was 0.84, with a range of 0.22–1.00. Sixty-eight per cent reported that they were 'extremely' or 'very satisfied' with their care and support, while a further 24% were 'quite satisfied'.

Table 3 shows overall quality of life and SCRQoL for the sample, both overall and for each type of SL. Those using long-term SL rated their quality of life more highly than those using day support or outreach, and also had better SCRQoL (numbers of those using respite were too small for differences to reach significance). The different aspects of SCRQoL were examined in terms of the percentage of people reporting the ideal situation or having no unmet need in that area of life. Those using long-term/residential SL support were more likely to report having all needs met for control over daily life, social participation, occupation and dignity. In contrast, those using outreach support were less likely to report having all needs met in these areas. Those using day support were less likely to report having all needs met in the areas of occupation and food and drink; those using respite/short breaks were also less likely to report having needs met in the food and drink domain.

In order to discover whether the easy-read element of the sample (many of whom were likely to have had a learning disability) was biasing these results, tests were re-run without this group. Outcomes for people using long-term SL were still statistically better for this smaller sample.

However, as described above, people using different types of SL support differ in terms of age, health and dependency and it is likely that this has an effect on their quality of life. In addition, people using long-term/residential SL receive more of the support, which may also have an influence. To allow for this, multiple regression analyses were conducted to facilitate assessment of the unique effect of each variable on SCRQoL and on overall quality of life. The models analysed 123 and 116 cases respectively (the remainder were excluded through casewise deletion due to

	Long term n (%)	Day support n (%)	Respite/short breaks n (%)	Outreach n (%)	Missing	Test	Ρ
Age							
Range	65–93	65–96	68–91	66–102			
Mean (SD)	71.6 (5.81)	78.3 (9.49)	77.9 (9.13)	81.3 (7.79)	10	F (3,136) = 13.65	<0.001*
Gender							
Female	26 (51.0)	19 (57.6)	2 (25.0)	24 (47.1)	7		0.419 <sup>†</sup>
Male	25 (49.0)	14 (42.4)	6 (75.0)	27 (52.9)			
Dependency (ADL score) <sup>‡</sup>							
Range	0–7	0–16	2–7	0–16			
Mean (SD)	1.84 (1.85)	5.44 (4.12)	4.43 (2.07)	7.23 (4.26)	14	F (3,132) = 19.27	<0.001*
Pain/discomfort <sup>§</sup>							
Mean (SD)	1.44 (0.54)	1.76 (0.65)	1.85 (0.90)	1.94 (0.53)	3		
None	30 (57.7)	12 (35.3)	3 (42.9)	9 (16.7)		χ²(3) = 19.03 <sup>¶</sup>	< 0.001
Moderate	21 (40.4)	18 (52.9)	2 (28.6)	39 (72.2)			
Extreme	1 (1.9)	4 (11.8)	2 (28.6)	6 (11.1)			
Anxiety/depression§							
Mean (SD)	1.33 (0.51)	1.68 (0.53)	1.75 (0.46)	1.94 (0.61)	4		
None	36 (69.2)	12 (35.3)	2 (25.0)	16 (30.8)		χ²(3) = 18.91 <sup>¶</sup>	0.001
Moderate	15 (28.9)	21 (61.8)	6 (75.0)	31 (59.6)			
Extreme	1 (1.9)	1 (2.9)	0 (0.0)	5 (9.6)			
Self-perceived health							
Mean (SD)	1.90 (0.85)	2.65 (0.77)	3.50 (1.07)	3.02 (0.91)	3	F (3,143) = 18.16	< 0.001
Very good	19 (36.5)	3 (8.8)	0 (0.0)	1 (1.9)			
Good	21 (40.4)	9 (26.5)	2 (25.0)	13 (24.9)			
Alright	10 (19.2)	19 (55.9)	1 (12.5)	28 (52.8)			
Bad	2 (3.9)	3 (8.8)	4 (50.0)	6 (11.3)			
Very bad	0 (0.0)	0 (0.0)	1 (12.5)	5 (9.4)			
Length of time using SL							
<1 year	2 (4.1)	4 (13.8)	1 (12.5	20 (41.7)		$\chi^2(3) = 52.92^{\circ}$	<0.001
>1 year but <3 years	5 (10.2)	13 (44.8)	2 (25.0)	20 (41.7)			
>3 years but <5 years	9 (18.4)	3 (10.3)	2 (25.0)	6 (12.5)			
>5 years but <10 years	16 (32.7)	7 (24.1)	0 (0.0)	2 (4.2)			
>10 years but <15 years	7 (14.3)	0 (0.0)	0 (0.0)	0 (0.0)			
>15 years but <20 years	2 (4.1)	1 (3.5)	1 (12.5)	0 (0.0)			
20 years or more	8 (16.3)	1 (3.5)	2 (25.0)	0 (0.0)			
Help with survey completion	n						
No help	9 (17.3)	6 (18.8)	1 (14.3)	2 (3.8)	6		0.001 <sup>†</sup>
Assistance	43 (82.7)	20 (62.5)	5 (71.4)	41 (77.4)			
Proxy	0 (0.0)	6 (18.8)	1 (14.3)	10 (18.9)			

Table 2 Characteristics of older people using different types of SL support

\*Bartlett's test of equal variances was statistically significant.

<sup>†</sup>Fisher's exact test was used, therefore no accompanying test statistic.

<sup>‡</sup>ADL = activities of daily living. 0–16, higher scores indicate greater dependency.

<sup>§</sup>0–3, higher scores indicate greater depression/ anxiety, greater pain/discomfort.

<sup>¶</sup>Kruskall-Wallis test used.

missing data on one or more variables). Table 4 shows the results for overall quality of life and SCRQoL. The models explained 37% of the variance in QoL scores and 45% of the variance in SCRQoL scores. Those who rated their health more poorly also reported a worse quality of life, as did those who experienced extreme anxiety or depression. Poorer SCRQoL was linked to reporting moderate pain or discomfort. Better SCRQoL was associated with being

younger and with being female. Once these factors were controlled for, the type of SL support did not have a significant effect on QoL or SCRQoL scores.

# Comparison with older people using alternative care and support

A key aim of the survey was to facilitate comparison of outcomes of older SL users to older users of other

	Long term	Day support	Respite/short breaks	Outreach	Overall	Test	Ρ
Overall quality of life score (1-	-5)*						
Mean	1.42	2.06	2.29	2.37	1.95	F (3,141) = 12.29	< 0.001
Standard deviation	0.64	0.89	0.76	0.93	0.91		
Range	1–3	1–4	1–3	1–5	1–5		
Missing	0	1	1	3	5		
SCRQoL score, weighted (-0.	.17 to 1.00) <sup>†</sup>						
Mean	0.95	0.78	0.80	0.77	0.84	<i>F</i> (3,131) = 12.32 <sup>‡</sup>	< 0.001
Standard deviation	0.07	0.21	0.16	0.18	0.18		
Range	0.67-1.00	0.22-1.00	0.55-0.98	0.29-1.00	0.22-1.00		
Missing	4	3	2	6	15		
Aspects of SCRQoL (% ideal/	no need)						
Control	98.1	76.5	87.5	65.5	80.5		<0.001 <sup>§</sup>
Social participation	86.0	65.7	85.7	63.0	72.6		0.029 <sup>§</sup>
Occupation	96.1	51.4	71.4	46.3	66.0		<0.001 <sup>§</sup>
Food and drink	100.0	88.6	75.0	98.2	95.3		0.002 <sup>§</sup>
Personal cleanliness and comfort	98.0	94.3	100.0	98.2	97.3		0.645 <sup>§</sup>
Accommodation cleanliness and comfort	100.0	94.3	100.0	100.0	98.7		0.157 <sup>§</sup>
Safety	98.0	88.6	85.7	85.5	90.5		0.081 <sup>§</sup>
Dignity	100.0	87.9	100.00	79.6	90.1		0.005 <sup>§</sup>

Table 3 Outcomes for older people using Shared Lives

\*Higher scores indicate poorer QoL.

<sup>†</sup>Higher scores indicate better SCRQoL.

<sup>‡</sup>Bartlett's test of equal variances was statistically significant.

<sup>§</sup>Fisher's exact test was used, therefore no accompanying test statistic.

social care services using data from the ASCS. The ASCS 2012 data set contains 164,569 cases. In order to reflect older people using SL, excluded were anyone: under 65; receiving nursing care; receiving equipment services from the LA but no other services; and those without information on outcomes. This resulted in a sample of over 30,000. Tests for differences between this subset of the ASCS 2012 and the SL data set indicated that the two samples differed on various background variables. There was a smaller proportion of females in the SL data set  $(\chi^2(1) = 22.45, P < 0.001)$ . The SL sample were also younger, with a greater proportion of people aged between 65 and 74 and fewer over 85 ( $\chi^2(2) = 63.88$ , P < 0.001), and less dependent (t(28530) = -3.20, P = 0.0014). They experienced less pain or discomfort  $(\chi^2(2) = 8.92, P = 0.012)$  and rated their health more favourably (z = -3.317, P < 0.001).

To adjust for these differences, PSM was used to generate a matched sample and facilitate an unbiased comparison of outcomes. The conditional probability (propensity score) that any individual in the two samples (ASCS or SL) might be allocated to SL was estimated using logistic regression. The choice of conditioning variables in the logistic regression model is best informed by prior evidence and theory (Guo *et al.* 2006, Austin 2011); in this case, however, a lack of prior research evidence in the area and the small number of available background variables meant that choice was limited.

A number of models were tested to determine the most appropriate predictors of the probability of being allocated to SL. The aim was to achieve the best possible predictive value but minimise the exclusion of SL cases (a number of variables had missing values and thus cases were excluded). The predictive ability of all models tested was poor for SL cases, due in part to the small proportion that SL formed of the overall sample; however, this does not invalidate the match achieved. The final model included gender, age group, experience of pain, ability to get in and out of bed, ability to deal with paperwork, and overall health. PSM was conducted using STATA/ PSMATCH2 based on nearest neighbour matching with calliper (1/4 SD) and without replacement (following Austin 2011). This produced a matched sample of 121 cases from the SL sample and 121 from the ASCS sample.

The matched SL sample was similar to the overall SL sample, differing only in terms of one indicator of dependency: ability to feed themselves (the nonmatched group were more likely to have difficulty

	Overall quality of life*			SCRQoL <sup>†</sup>			
	Coefficient (Beta)	95% CI	P-value	Coefficient (Beta)	95% CI	P-value	
Gender <sup>‡</sup>	0.13 (0.07)	-0.40 to 0.14	0.334	0.07 (0.18)	0.00 to 0.13	0.037	
Age	0.01 (0.05)	-0.01 to 0.03	0.617	-0.01 (-0.23)	-0.01 to 0.00	0.053	
ADL score	-0.02 (-0.08)	-0.06 to 0.03	0.446	-0.01 (-0.11)	-0.01 to 0.00	0.298	
Overall health	0.29 (0.31)	0.10 to 0.48	0.003	-0.03 (0.15)	-0.07 to 0.01	0.165	
Pain/discomfort <sup>§</sup>							
Moderate	0.16 (0.09)	-0.12 to 0.44	0.249	-0.09 (-0.24)	-0.15 to -0.02	0.009	
Extreme	-0.17 (-0.05)	-0.71 to 0.37	0.525	-0.02 (-0.02)	-0.12 to 0.08	0.726	
Anxiety/depression <sup>¶</sup>	× ,						
Moderate	0.21 (0.23)	-0.11 to 0.53	0.194	-0.02 (-0.06)	-0.10 to 0.06	0.609	
Extreme	1.04 (0.23)	0.12 to 1.95	0.026	-0.16 (-0.18)	-0.35 to 0.02	0.084	
Easy-read version**	-0.08 (-0.04)	-0.42 to 0.26	0.631	-0.00 (-0.01)	-0.06 to 0.05	0.930	
SL type <sup>††</sup>	× ,						
Day support	0.25 (0.12)	-0.18 to 0.68	0.258	-0.08 (-0.19)	-0.17 to 0.00	0.064	
Respite/short breaks	0.25 (0.07)	-0.33 to 0.83	0.395	-0.02 (-0.02)	-0.14 to 0.11	0.773	
Outreach	0.40 (0.21)	-0.73 to 2.14	0.118	-0.03 (-0.09)	-0.12 to 0.05	0.438	
Overall model significance	F(12, 110) = 6.56,			F(12,103) = 9.93,			
3	<i>P</i> < 0.001			<i>P</i> < 0.001			
R <sup>2</sup>	0.37			0.45			

Table 4 Factors associate	ed with	variations	in	outcome
---------------------------	---------	------------	----	---------

\*Higher scores indicate poorer QoL.

<sup>†</sup>Higher scores indicate better SCRQoL.

 $^{\ddagger}$ Male = 1, female = 0.

Base = no pain/discomfort.

<sup>¶</sup>Base = no anxiety/depression.

\*\*Easy-read = 1, main version = 0.

<sup>††</sup>Base = long term/residential.

with this task, P = 0.046, Fisher's exact). The matched ASCS sample included people using a mix of social care services including home care, day support and residential care, reflecting the varied nature of support available through SL. As they now 'matched' the SL users, this sample of the ASCS was on average younger ( $\chi^2(2) = 57.35$ , P < 0.001), less dependent (t(117.362) = -4.26, P < 0.001), in better health (z = -3.284, P = 0.001) and experienced less pain and discomfort than the overall sample ( $\chi^2(2) = 8.74$ , P = 0.01). There was a greater proportion of men ( $\chi^2(1) = 21.21$ , P < 0.001) and of people who completed the easy-read questionnaire ( $\chi^2(1) = 18.13$ , P < 0.001) in the matched sample.

After matching, there was balance between the SL and ASCS samples on all the characteristics entered as variables in the model. However, there were significant differences between the groups in terms of the source of help received to complete the questionnaire. In addition, more people in the SL sample completed the easy-read questionnaire ( $\chi^2(1) = 12.58$ , P < 0.001). It would not have been appropriate to use these variables in the logistic regression model, as they were not true background variables, but a feature of the questionnaire.

Table 5 shows the outcomes for the matched groups. There was a small but statistically significant difference in overall quality of life with SL users reporting better quality of life than those in the comparison group. However, there was no significant difference in overall SCRQoL scores for the two groups. For the individual domains of SCRQoL, the SL sample contained more people reporting the 'ideal' situation than simply having 'no needs' in the domains of food and drink and accommodation cleanliness. Results in the control, social participation and occupation domains were also in this direction, although did not reach statistical significance. Results in the safety domain, also non-significant, showed a different pattern, with more people from the ASCS sample reporting the 'ideal' situation.

#### Discussion

The aim of this paper was to describe the outcomes of older people using SL support and to compare these to those of older people using alternative services, addressing the lack of evidence available about the potential of SL for older people. Although there are a number of caveats to our findings, the research

	Shared Lives	ASCS sample		
	n (%)	n (%)	Test statistic	P
Overall quality of life score (1-5)				
Mean	1.94	2.18	<i>z</i> = 1.96	0.05
Standard deviation	0.91	0.97		
Range	1–5	1–5		
SCRQoL				
Mean	0.84	0.82	t(240) = -0.4856	0.63
Standard deviation	0.18	0.18		
Range	0.22-1.00	0.05–1.00		
Control				
No needs – ideal state	51 (42.2)	41 (33.9)	$\gamma^2(3) = 4.40$	0.22
No needs	51 (42.2)	49 (40.5)	χ (σ) πτο	0.22
Some needs	16 (13.2)	28 (23.1)		
High needs	3 (2.5)	3 (2 5)		
Social participation	0 (2.0)	0 (2.0)		
No needs – ideal state	51 (42 2)	48 (39 7)	$\gamma^2(3) = 1.32$	0.72
No needs	10 (34 7)	49 (40 5)	$\chi(0) = 1.02$	0.72
Some needs	42(04.7)	10 (15 7)		
High needs	24 (19.0) A (3.3)	5 (4 1)		
	4 (0.0)	5 (4.1)		
No poods ideal state	54 (44 6)	41 (22 0)	$x^{2}(2) = 5.28$	0.15
No needo	04 (44.0) 20 (24.9)	41 (33.9)	$\chi$ (3) – 5.38	0.15
Some noode	30 (24.0) 32 (26.5)	40 (07.2)		
High poods	52(20.5)	7 (5.9)		
Diamity	5 (4.1)	7 (5.8)		
Dignity	70 (E0 E)		$x^{2}(2) = 4.00$	0.04
No needs – lueal state	72 (39.5)	67 (55.4)	$\chi$ (3) = 4.20	0.24
No needs	36 (29.8)	46 (38)		
Some needs	13 (10.7)	7 (5.8)		
High needs	0	1 (0.8)		
Safety	00 (00 0)		2(0) 0.47	0.40
No needs – Ideal state	83 (68.6)	89 (73.6)	$\chi^{2}(3) = 2.47$	0.48
No needs	27 (22.3)	27 (22.3)		
Some needs	9 (7.4)	4 (3.3)		
High needs	2 (1.7)	1 (0.8)		
Food and drink	00 (70 0)		2(0) 0.74	
No needs – ideal state	96 (79.3)	79 (65.3)	$\chi^{2}(3) = 8.74$	0.03
No needs	20 (16.5)	38 (31.4)		
Some needs	5 (4.1)	3 (2.5)		
High needs	0	1 (0.8)		
Accommodation			2	
No needs – ideal state	96 (79.3)	80 (66.1)	$\chi^{2}(3) = 5.94$	0.05
No needs	23 (19)	35 (28.9)		
Some needs	2 (1.7)	6 (5)		
High needs	0	0		
Personal cleanliness and comfort			0	
No needs – ideal state	74 (61.2)	74 (61.2)	$\chi^2(3) = 1.11$	0.78
No needs	44 (36.4)	41 (33.9)		
Some needs	2 (1.7)	4 (3.3)		
High needs	1 (0.8)	2 (1.7)		

Table 5 Comparison of outcomes between SL and ASCS samples

offers some useful messages and provides a starting point for developing the evidence base for SL.

The study provides evidence that SL can deliver good outcomes, particularly for overall quality of life, echoing and building on the positive findings from more general research on SL (Fiedler 2005, NAAPS and IESE, 2009; Shared Lives Plus, 2014). The majority of SL users rated their overall quality of life highly and the average score for SCRQoL was also high, at a similar level to that found among nonsocial care users (Netten *et al.*, 2012), with those using long-term placements reporting the best quality of life. For some respondents, the amount of support they received from SL would have been small – for example, a few hours a week for outreach support or a few weeks a year for short breaks. This means that the effect of SL on some aspects of their quality of life is likely to be minimal and outcomes may reflect other aspects of support than SL. Nonetheless, people using long-term placements had consistently good outcomes across all domains of SCRQoL and it is these people where outcomes are most likely to reflect the support from SL. However, results of the regression models did suggest that it may be the differing characteristics of individuals using different models of SL, rather than the type of support itself, that was affecting quality of life outcomes.

There was some evidence that people using SL report a better quality of life than a similar group of older people using alternative care and support services. Although there was no difference in overall SCRQoL scores, 'food and drink' and 'accommodation comfort and cleanliness' were aspects of SCRQoL which appeared to be better for the SL users, with more people reporting the 'ideal' situation where needs are met to the optimum level. Results in some of the other domains including control, social participation and occupation also showed this pattern, although were not statistically significant. Responses to open questions included in the survey (described in Brookes et al. 2016) appeared to support these findings, with users of all types of SL support reporting that they valued the increased opportunities for social contact and getting involved in activities, many describing their SL carer as a friend or source of company and the value of feeling 'part of a family'. Having choice over daily activities was seen as a key benefit of SL, as was the help to regain or maintain independence, both important elements of having control over daily life. It may be that the SL model of support could help address some of the negative aspects associated with older age such as loneliness and decreased social opportunities (Windle et al. 2011), and promote a sense of control. Indeed, it has been suggested that, for older people, control may be best achieved through relationships of trust with those that support them (Woolham 2015), something which SL services would seem ideally placed to facilitate.

The finding that users of SL rated their overall quality of life more highly than those using alternative care and support services, while there was no difference in SCRQoL, may indicate that SL offers some additional benefits not easily captured by the ASCOT measure, such as emotional support. Again, respondents' comments about SL lend some support to this suggestion, linking the emotional support provided by the SL carer directly to improvement or maintenance of mental health and well-being (Brookes *et al.* 2016).

#### Limitations

There are a number of limitations to the analysis which need to be borne in mind when interpreting the results. The sample size was small, representing approximately 9% of older people using SL support, although it is the largest sample of older SL users achieved in research to date. Given that there is scarce information on the characteristics of SL users, it is difficult to determine how representative the sample is of the population, and care needs to be taken in generalising the findings. However, SL is generally considered to be suited to older people at the more able end of the spectrum who do not need nursing-level care, and the characteristics of the sample in this study would seem to reflect that.

The sample was also complex: four types of SL support were represented, with users of each type having different characteristics. As there is so little evidence on SL, it was deemed important for the study to be as inclusive as possible but the small numbers in each subgroup meant more complex analysis was not possible; it would have been useful, for example to have considered the 'outreach' group separately given that this model differs from 'traditional' SL, and came only from one scheme. It was also not possible (given the small numbers) to analyse results by scheme; again a limitation to the findings presented here.

There are also limitations to the analysis used to compare the ASCS and SL samples. In the absence of a 'real world' comparison group, statistical matching techniques offer an alternative. However, the comparison made can only apply to the matched sample, not to those for whom there was no 'match' – in this case, the matched sample represents a tiny proportion of older people using social care and support services in 2011/2012. Although the older people drawn from the ASCS data set were using a range of types of care and support, they were younger, relatively able and in better health compared to the overall sample of older people. More information is needed about the characteristics of the older people using SL in order to be confident that this was comparing like with like.

### Conclusion

The research presented suggests that SL can provide a viable option alongside other forms of care and support for older people and one which reflects a number of current policy initiatives, offering choice and control and personalised support in a community setting (DH 2010, 2012, Care Act 2014). However, it is important to note that, although the sector is expanding, SL forms only a small part of social care and support provision for older people. Although there is a support for expansion of provision (Brookes & Callaghan, 2013), effort is needed to facilitate growth and raise awareness of the potential of SL. Further research to collect information on outcomes from a larger sample of older people would enable greater generalisability of findings, more sophisticated analysis of the different types of SL support and further comparative analysis.

#### Acknowledgements

This article draws on independent research funded by the National Institute for Health Research (NIHR) School for Social Care Research (SSCR), England. The views expressed are those of the authors and not necessarily those of the NIHR SSCR, the Department of Health or National Health Service, England. Acknowledgement is due to the service users and carers who gave their time and shared their experiences and to Shared Lives Plus and the participating Shared Lives schemes for supporting the research. Acknowledgement is also due to the local and overall project advisory groups (comprising both professional and lay members) for helpful advice throughout the research process, to Dr Stephen Allan for statistical advice and to Professor Ann Netten and two anonymous peer reviewers for helpful comments on an earlier draft.

#### References

- Austin P.C. (2011) An introduction to propensity score methods for reducing the effects of confounding in observational studies. *Multivariate Behavioral Research* **46** (3), 399–424.
- Bell J. & Litherland R. (2013) Shared Lives and Dementia: Final Report of the National Shared Lives and Dementia Project 2010-2013. Shared Lives South West and Innovations in Dementia, Devon, UK.
- Blood I. (2013) A Better Life: Valuing Our Later Years. Joseph Rowntree Foundation, York, UK.
- Brookes N. & Callaghan L. (2013) What next for Shared Lives? Family-based support as a potential option for older people. *Journal of Care Services Management* 7 (3), 87–94.
- Brookes N., Palmer S. & Callaghan L. (2016) 'I live with other people and not alone': a survey of the views and experiences of older people using Shared Lives (adult placement). *Working with Older People* **20** (3), 179–186.
- Bowling A. (1995) *Measuring Disease*. Open University Press, Buckingham, UK.
- Callaghan L., Netten A., Brookes N., Fox D. (2012) Personalisation of Services Scoping Study: Research Findings. NIHR School for Social Care Research, London, UK. Available

at: http://sscr.nihr.ac.uk/PDF/Findings/Findings\_2\_per sonalisation\_web.pdf (accessed on 24/03/2014).

- Care Act (2014) (*c*.23). The Stationary Office, London, UK. Available at: www.legislation.gov.uk/ukpga/2014/23/ contents/enacted (accessed on 02/04/2015).
- Care Quality Commission (2014) *The State of Health Care and Adult Social Care in England 2013/14*. The Stationery Office, London, UK.
- Chen X., Ender P., Mitchell M. & Wells C. (2003). *Regression with Stata*, available at: http://www.ats.ucla.edu/stat/sta ta/webbooks/reg/default.htm (assessed on 01/04/2014)
- Dagnan D. (1997) Family placement schemes offering long term care for adults with learning disabilities: a review of the evaluation literature. *Disability and Society* **12** (4), 593–604.
- Dagnan D. & Drewett R. (1988) Community based care for people with a mental handicap: a family placement scheme in County Durham. *British Journal of Social Work* 18 (6), 543–575.
- DH (2010) A Vision for Adult Social Care: Capable Communities and Active Citizens. Department of Health, London, UK.
- DH (2012) Caring For Our Future. The Stationery Office, London, UK.
- Equality and Human Rights Commission (2011) *Close to Home: An Inquiry Into Older People and Human Rights in Home Care.* Equality and Human Rights Commission, London, UK.
- Fiedler B. (2005) Adult Placement in England: A Synthesis of the Literature. Social Care Institute for Excellence, London, UK.
- Fox A. (2011) A new model for care and support: sharing lives and taking charge. *Working with Older People* **15** (2), 58–63.
- Gilleard C., Hyde M. & Higgs P. (2007) The impact of age, place, aging in place and attachment to place on the wellbeing of the over 50s in England. *Research on Aging* **29** (6), 590–605.
- Guo S., Barth R.P. & Gibbons C. (2006) Propensity score matching strategies for evaluating substance abuse services for child welfare clients. *Children and Youth Services Review* 28 (4), 357–383.
- Lawton M.P. (1985) The elderly in context: perspectives from environmental psychology and gerontology. *Environment and Behaviour* **17** (4), 501–519.
- McConkey R., McConaghie J., Roberts P. & King D. (2002) *Adult Family Placement Schemes for Family Carers*. University of Ulster, Ulster, UK.
- NAAPS (2010) *The Shared Life of My Choice*. NAAPS, London, UK. Available at: http://sharedlivesplus.org.uk/images/ publications/SL-of-my-Choice-commissioners-and-provide rs.pdf (accessed on 07/04/2015).
- NAAPS and IESE (2009) A Business Case for Shared Lives. NAAPS, London, UK.
- Netten A., Burge P., Malley J. *et al.* (2012) Outcomes of social care for adults: Developing a preference weighted measure. *Health Technology Assessment* **16** (16), 1–165.
- Newbronner L., Chamberlain R., Bosanquet K., Bartlett C., Sass B. & Glendinning C. (2011) *Keeping Personal Budgets Personal: Learning from the Experiences of Older People, People with Mental Health Problems and Their Carers.* SCIE Report 40. Social Care Institute for Excellence, London, UK.
- Potoglou D., Burge P., Flynn T., Netten A., Malley J., Forder J. & Brazier J. (2011) Best-worst scaling vs discrete choice experiments: an empirical comparison using social care. *Social Science and Medicine* **72** (10), 1717–1727.

- Robine J.M., Jagger C. & Romieu I. (2002) Selection of a Coherent Set of Health Indicators for the European Union. Phase 2: Final Report. Euro-REVES, Montpellier, France.
- Robinson C. & Simons K. (1996) In Safe Hands? Quality and Regulation in Adult Placements for People with Learning Difficulties. Joint Unit for Social Services Research, University of Sheffield, Sheffield, UK.
- Rosenbaum P.R. & Rubin D.B. (1983) The central role of the propensity score in observational studies for causal effects. *Biometrika* **70** (1), 41–55.
- Schofield P. (2009) Adult Placements in Mental Health Care in England and Wales: An Exploratory Study. PhD Thesis. University of London, London, UK.
- Shared Lives Plus (2014) *The State of Shared Lives in England*. Shared Lives Plus, Liverpool, UK.
- Shared Lives Plus (2016) *The State of Shared Lives in England*. Shared Lives Plus, Liverpool, UK.

- Tabachnick B.G. & Fidell L.S. (2007) Using Multivariate Statistics, 5<sup>th</sup> edn. Allyn and Bacon, Boston, MA, USA.
- Valios N. (2010) Good Practice: time to share scheme for dementia patients. Community Care Online. Available at: http://www. communitycare.co.uk/2010/10/28/good-practice-time-to-sha re-scheme-for-dementia-patients/ (accessed on 07/04/2015).
- Ware P. (1987) Advantages and disadvantages of boarding out care for elderly people. *Adult Foster Care Journal* 1 (3), 141–159.
- Windle K., Francis J. & Coomber C. (2011) Research Briefing 39: Preventing Loneliness and Social Isolation: Interventions and Outcomes. Social Care Institute for Excellence, London, UK.
- Woolham J. (2015) Are Personal Budgets Always the Best Way of Delivering Personalised Social Care to Older People? Research Findings. NIHR School for Social Care Research, London, UK. Available at: http://www.sscr.nihr.ac.uk/ PDF/Findings/RF19.pdf (accessed on 19/07/2016).